Author's response to reviews

Title: Clavicular stress fracture in a cricket fast bowler: A case report

Authors: Jeremy A Read (jezread@gmail.com)  Phillip Bell (bellph@bupa.com)

Version: 5 Date: 16 April 2008

Author's response to reviews:

Point by point resubmission

Reviewer 1 (John Harrison)

Point 1. Spelling of clavicle checked and changed
Point 2. CT scan image now included showing location of fracture in clavicle in the region of the coracoclavicular ligament

Reviewer 2 (Joseph Pooley)

Point 1. Distal replaced with lateral in relation to clavicle in the text
Point 2. CT scan image now included to show fracture
Point 3. "Presumptive" inserted before "boweling related shoulder ... "

Abstract has been adjusted to the required format and figure legends now after the references